Author's response to reviews

Title: Patterns in sedentary and exercise behavior and associations with overweight in 9-14-year-old boys and girls - a cross-sectional study-

Authors:
Saskia J te Velde (s.tevelde@erasmusmc.nl)
Ilse de Bourdeaudhuij (ilse.debourdeaudhuij@ugent.be)
Inga Thorsdottir (ingathor@landspitali.is)
Mette Rasmussen (M.Rasmussen@pubhealth.ku.dk)
Maria Hagstromer (mahag@biosci.ki.se)
Knut-Inge Klepp (k.i.klepp@medisin.uio.no)
Johannes Brug (j.brug@erasmusmc.nl)

Version: 5 Date: 25 January 2007

Author's response to reviews:

Dear Editor,

Thank you for your email dd 23 January 2007, mentioning that our manuscript is, in principle, accepted for publication in BMC Public Health.

We made the minor revisions according your suggestions. We changed the subheadings of the abstract and elaborated more on the context of the current study in the Background of the abstract.

We also renamed 'introduction' into 'Background' and we changed the title into 'Patterns in sedentary and exercise behaviors and associations with overweight in 9-14-year-old boys and girls - a cross-sectional study -', and adapted the lay-out of the tables, according the Authors' checklist saying that the title should include the type of intervention or research and table cell borders must have visible black lines.

We hope the manuscript is now ready for publication.

Sincerely yours,

On behalf of all authors, Saskia te Velde.